Author's response to reviews

Title: Male and Female Adult Population Health Status in China: A Cross-Sectional National Survey

Authors:

Jing Shi (shijing8989@yahoo.com.cn)
Meina Liu (liumeina@ems.hrblnu.edu.cn)
Qiuju Zhang (zqj81129@sina.com)
Mingshan Lu (Lu@ucalgary.ca)
Hude Quan (hquan@ucalgary.ca)

Version: 3  Date: 1 August 2008

Author's response to reviews: see over
Our reformatting changes are below:

Formatting changes requested
--------------------------------

Major revisions (we require authors to make these changes)

No abstract present in submission system. Please upload abstract (Please ensure that the Abstract is present in manuscript and matches exactly what is in the submission system.)

It is done.

Please reformat text (boldness) in the abstract section and throughout the manuscript

It is done

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].

It is done

Minor revisions (we can make these changes on the authors behalf)

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system

Done

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.

Done

In references 3 and 19, the website links would not load correctly
These two websites can be correctly loaded in China but not in Canada (I tried that in Canada). It may be temporarily blocked.

If a table runs over two pages the column headings must not be repeated. Please amend on table 1 and 3

Done

Additional file - Please delete the additional file that contains your response to the reviewers.
No additional file

Regards

Hude Quan